Author's response to reviews

Title: Serum under-O-glycosylated IgA level is not correlated with glomerular IgA deposition based upon heterogeneity in the composition of immune complexes in IgA nephropathy

Authors:

Kenji Satake (k-satake@juntendo.ac.jp)
Yoshio Shimizu (yosimizu@juntendo.ac.jp)
Yohei Sasaki (yhsasaki@juntendo.ac.jp)
Hiroyuki Yanagawa (yanagawa@juntendo.ac.jp)
Hitoshi Suzuki (shitoshi@juntendo.ac.jp)
Yusuke Suzuki (yusuke@juntendo.ac.jp)
Satoshi Horikoshi (satoshi@juntendo.ac.jp)
Shinichiro Honda (shonda@md.tsukuba.ac.jp)
Kazuko Shibuya (kazukos@md.tsukuba.ac.jp)
Akira Shibuya (ashibuya@md.tsukuba.ac.jp)
Yasuhiko Tomino (yasu@juntendo.ac.jp)

Version: 4  Date: 3 June 2014

Author's response to reviews: see over
Dear Dr. Henderson:

We respectfully re-submit the enclosed manuscript entitled, “Serum under-O-glycosylated IgA level is not correlated with glomerular IgA deposition based upon heterogeneity in the composition of immune complexes in IgA nephropathy.” We modified this manuscript according to editor’s suggestions.

The responses for the comments are described in the following. We hope this version is suitable publication in BMC nephrology.

Thank you, again for your consideration. We look forward to hearing from you.

Sincerely,

Kenji Satake, MD, Yoshio Shimizu, MD, PhD, and Yasuhiko Tomino, MD, PhD
Division of Nephrology, Department of Internal Medicine, Juntendo University
Faculty of Medicine, 2-1-1, Hongo, Bunkyo-ku, Tokyo 113-8421, Japan
k-satake@juntendo.ac.jp, yosimizu@juntendo.ac.jp, yasu@juntendo.ac.jp

Responses to editor’s suggestions

#1. Abstract

We corrected the subheading of the abstract to ‘Background, Methods, Results and Conclusions.’

#2. Authors Contributions

We changed each contributor’s name into their initials.

#3. Tables

We removed the Tables from the uploaded files and input all of them into main text.